Author's response to reviews

Title: Survey and online discussion groups to develop a patient-rated outcome measure on acceptability of treatment response in vitiligo

Authors:

Selina K Tour (selina.tour@nottingham.ac.uk)
Kim S Thomas (kim.thomas@nottingham.ac.uk)
Dawn-Marie Walker (dawn-marie.walker@nottingham.ac.uk)
Paul Leighton (paul.leighton@nottingham.ac.uk)
Adrian SW Yong (yongszewai@gmail.com)
Jonathan Batchelor (jonathan.batchelor@nottingham.ac.uk)

Version: 2
Date: 26 February 2014

Author's response to reviews: see over
RE: Pre-Review Request- MS: 1329999110120560

Dear Maria Merrie Jul Ladag,

The following amendments have been made to the revised manuscript submitted in response to comments received prior to peer-review.

1) **Please include a separate conclusions section after the discussion that contains the strengths and limitations subsection** – a conclusions section heading has now been added (pg.13); this includes the strengths and limitations of the study and the implications for future research.

2) **Consent to publish:** The wording of the figure 4 legend has now been amended (pg.20) to confirm that full consent was obtained for use of this image for this research and publication. Please let us know if you need us to provide a copy of the patient’s consent form.

3) **Authors Contributions:** Individual contribution has been detailed (pg.14) above the acknowledgements section of the manuscript to provide detail of individual contributions.

I hope that the revised manuscript meets your requirements to proceed to peer-review.

Kind regards,

Selina Tour

On behalf of the authors

Contact details: Selina.tour@nottingham.ac.uk